1. This study is very interesting. The authors evaluated the wound healing activity and the prediction of anti-diabetic foot ulcer mechanism based on component analysis of Ruyi Jinhuang powder. The conclusion is supported by the results. Well done study. Thank you.
   Thank you for your advice.

2. There are some minor language polishing which should be corrected. 2. The conclusion in the abstract is too long. Please short it. 3. The figures should be updated with high quality images. 4. The conclusion in the main text is missing. Please add it.
   Thank you for your advice. The manuscript has been re-submitted and retouched on the AJE system. We have updated the conclusion and the figures.

3. The abstract should be shorted. 2. Tables and figures should be moved to the end of the text. 3. References list should be updated.

   Thank you for your advice. The abstract has been shorted. Tables and figures have been moved to the end of the text. We have revised the references list.